\documentclass[11pt]{osh}
\usepackage{amsthm, amsmath, amssymb, amsbsy, amsfonts, latexsym, euscript,amscd}
\usepackage{calrsfs}
\usepackage{graphicx}


\newtheorem{theorem}{\bf \indent Theorem}[section]
\newtheorem{proposition}{\bf \indent Proposition}[section]
\newtheorem{lemma}{\bf \indent Lemma}[section]
\newtheorem{corollary}{\bf \indent Corollary}[section]

\theoremstyle{remark}
\newtheorem{remark}{\bf \indent Remark}[section]
\newtheorem{definition}{\bf \indent Definition}[section]
\newtheorem{example}{\bf \indent Example}[section]
\newtheorem{problem}{\bf \indent Problem}[section]
\numberwithin{equation}{section}


\setcounter{page}{67}
\begin{document}
\begin{thebibliography}{99}

\fontsize{10.6pt}{4mm}{\selectfont \small

\vskip3pt\bibitem{chet} R.\,G.~Chetaev, {\it The stability of motion}. New York, Pergamon Press, 1961.

\vskip3pt\bibitem{dzh1} J.\,K.~Hale, {\it Oscillations in nonlinear systems}. New York, Dover Publications, 1992.

\vskip3pt\bibitem{dzh2} J.\,K.~Hale, {\it Theory of functional differential equations}. New York Springer-Verlag, 1977.

\vskip3pt\bibitem{mas} J.\,L.~Massera, J.\,J.~Schaffer, {\it Linear differential equations and function spaces}. New York, Academic Press, 1966.

\vskip3pt\bibitem{yu} Z.~Shao, ''Global structure stability of Riemann solutions for linearly degenerate hyperbolic conservation laws under small BV perturbations of the initial data'', \textit{Nonlinear Analysis: Real World Applications} {\bf 11}, 3791--3808 (2010). https://doi.org/10.1016/j.nonrwa.2010.02.009.

\vskip3pt\bibitem{ka} Z.~Shao, ''Global weakly discontinuous solutions to the mixed initial-boundary value problem for quasilinear hyperbolic systems'', { \it Mathematical Models and Methods in Applied Sciences} {\bf 19}, 1099--1138 (2009).

\vskip3pt\bibitem{t1} M.~Benchohra, J.~Henderson, S.\,K.~Ntouyas, \emph{Impulsive differential equations and inclusions}, Contemporary Mathematics and Its Application, Hindawi Publishing Corporation, New York, 2006.

\vskip3pt\bibitem{t2} J.~Catlla, D.\,G.~Schaeffer, Th.\,P.~Witelski, E.\,E.~Monson, A.\,L.~Lin, ''On spiking models for synaptic activity and impulsive differential  equations'', {\it SIAM Review} {\bf 50} (3), 553--569 (2008).

\vskip3pt\bibitem{t3} A.~Halanay, D.~Veksler, \emph{Qualitative theory of impulsive systems}. Mir, Moscow, 1971. [in Russian]

\vskip3pt\bibitem{t4} V.~Lakshmikantham, D.\,D.~Bainov, P.\,S.~Simeonov, \emph{Theory of impulsive differential equations}, World Scientific, Singapore, 1989.

\vskip3pt\bibitem{t5} N.\,A.~Perestyk, V.\,A.~Plotnikov, A.\,M.~Samoilenko, N.\,V.~Skripnik, \emph{Differential equations with impulse Effect: Multivalued Right-Hand Sides with Discontinuities}, DeGruyter Stud. {\bf 40}, Math.  Walter de Gruter Co., Berlin, 2011.

\vskip3pt\bibitem{t6} A.\,M.~Samoilenko, N.\,A.~Perestyk, \emph{Impulsive differential equations}, World Science, Singapore, 1995.

\vskip3pt\bibitem{t7} I.~Stamova, G.~Stamov, \emph{Impulsive biological models}, In Applied impulsive mathematical models, CMS Books in Mathematics, Springer, Cham., 2016.

\vskip3pt\bibitem{t8} A.\,N.~Abdullozhonova, T.\,K.~Yuldashev, A.\,K.~Fayziyev, ''Mixed problem for an impulsive parabolic integro-dfferential equation with involution and nonlinear conditions'', {\it Lobachevskii Journal of Mathematics} \textbf{45} (3), 899--911 (2024). https://doi.org/10.1134/S199508022460078X

\vskip3pt\bibitem{t9} A.~Anguraj, M.\,M.~Arjunan, ''Existence and uniqueness of mild and classical solutions of impulsive evolution equations'', {\it Electronic  Journal of Differential Equations} {\bf 2005} (111), 1--8 (2005).

\vskip3pt\bibitem{t10} A.~Ashyralyev, Ya.\,A.~Sharifov, ''Existence and uniqueness of solutions for nonlinear impulsive differential equations with two-point and integral boundary  conditions'', {\it Advances in Difference Equations}  {\bf 2013} (173), (2013).

\vskip3pt\bibitem{ass1} A.\,T.~Assanova, M.\,A.~Mukash, O.\,M.~Stanzhytskyi, ''A problem with impulse actions for nonlinear ODEs'', {\it International Journal of Mathematics and Physics} {\bf 14} (1), 23--31 (2023).

\vskip3pt\bibitem{t11} Ch.~Bai, D.~Yang, ''Existence of solutions for second-order nonlinear impulsive differential equations with periodic boundary value  conditions'', {\it Boundary Value Problems (Hindawi Publishing Corporation)} {\bf 2007} (41589), 1--13 (2007).

\vskip3pt\bibitem{t12} L.~Bin, L.~Xinzhi, L.~Xiaoxin, ''Robust global exponential stability of uncertain impulsive systems'', {\t Acta Mathematika Scientia} {\bf 25} (1), 161--169 (2005).

\vskip3pt\bibitem{t13} A.\,K.~Fayziyev, A.\,N.~Abdullozhonova, T.\,K.~Yuldashev, ''Inverse problem for Whitham type multi-dimensional differential equation with impulse effects'', {\it Lobachevskii Journal of Mathematics} \textbf{44} (2), 570--579 (2023).

\vskip3pt\bibitem{ass2} A.\,E.~Imanchiyev, A.\,T.~Assanova, A.~Molybaikyzy, ''Properties of a nonlocal problem for hyperbolic equations with impulse discrete memory'', {\it Lobachevskii Journal of Mathematics} {\bf 44} (10), 4299--4309 (2023).

\vskip3pt\bibitem{t14} M.\,J.~Mardanov, Ya.\,A.~Sharifov, M.\,H.~Habib, ''Existence and uniqueness of solutions for first-order nonlinear differential equations with two-point and integral boundary conditions'', {\it Electrnic Journal of Differential Equations} {\bf 2014} (259), 1--8 (2014).

\vskip3pt\bibitem{ass3} S.~Mynbayeva, A.~Assanova,  R.\,A.~Uteshova, ''Solution to a linear boundary value problem for an impulsive integro-differential equation'', {\it Differential Equations and Dynamical Systems} 1--18 (2023). https://doi.org/10.1007/s12591-023-00665-8

\vskip3pt\bibitem{t15} T.\,K.~Yuldashev, A.\,K.~Fayziyev, ''On a nonlinear impulsive system of integro-differential equations with degenerate kernel  and maxima'', {\it Nanosystems: Physics. Chemistry. Mathematics}. {\bf 13} (1), 36--44 (2022).

\vskip3pt\bibitem{t16} T.\,K.~Yuldashev, A.\,K.~Fayziyev, ''Integral condition with nonlinear kernel for an impulsive system of differential equations with maxima  and redefinition vector'', {\it Lobachevskii Journal of Mathematics} {\bf 43} (8), 2332--2340 (2022).

\vskip3pt\bibitem{t17} T.\,K.~Yuldashev, A.\,K.~Fayziyev, ''Inverse problem for a second order impulsive system of integro-differential equations with two redefinition vectors  and mixed maxima'', {\it Nanosystems: Physics. Chemistry. Mathematics} \textbf{14} (1), 13--21 (2023). https://doi.org/10.17586/2220-8054-2023-14-1-13-21

\vskip3pt\bibitem{t21} T.\,K.~Yuldashev, Kh.\,Kh.~Saburov, T.\,A.~Abduvahobov, ''Nonlocal problem for a nonlinear system of fractional order impulsive integro-differential equations with  maxima'', {\it Chelyabinsk Physical-Mathematical Journal} \textbf{7} (1), 113--122 (2022).

\vskip3pt\bibitem{tt1} V.\,M.~Abdullayev, ''Numerical solution to optimal control problems with multipoint and integral conditions'', {\it Proceedings of the Institute of Mathematics and Mechanics} {\bf 44} (2), 171--186 (2018).
		
\vskip3pt\bibitem{tt2} S.~Albeverio, Sh.\,A.~Alimov, ''On a time-optimal control problem associated with the heat exchange process'', {\it Applied Mathematics and Optimization} {\bf 57} (1), 58--68 (2008).
		
\vskip3pt\bibitem{tt3} B.\,Y.~Ashirbaev, T.\,K.~Yuldashev, ''Derivation of a controllability criteria for a linear singularly perturbed discrete system with small step'', {\it Lobachevskii Journal of Mathematics} {\bf 45} (3), 938--948 (2024).
		
\vskip3pt\bibitem{tt4} A.\,A.~Azamov, G.\,I.~Ibragimov, K.~Mamayusupov, M.\,B.~Ruzibaev, ''On the stability and null controllability of an infinite system of linear differential equations'', {\it Journal of Dynamical and Control Systems}  (2021). https://doi.org/10.1007/s10883-021-09587-6
		
\vskip3pt\bibitem{tt5} A.\,A.~Azamov, M.\,B.~Ruziboev, ''The time-optimal problem for evolutionary partial differential equations'', {\it Journal of Applied Mathematics and Mechanics} {\bf 77} (2), 220--224 (2013).

\vskip3pt\bibitem{tt6} A.\,G.~Butkovsky, L.\,M.~Pustylnikov, \emph{Theory of mobile control of systems with distributed parameters}, [in Russian] Nauka, Moscow, 1980.
		
\vskip3pt\bibitem{tt7} V.\,S.~Deineka, ''Optimal control of the dynamic state of a thin compound plate'', {\it Cybernetics and Systems Analysis} {\bf 42} (4), 151--157 (2006).
		
\vskip3pt\bibitem{tt8} R.~Deumlich, K.\,H.~Elster, ''Duality theorems and optimality, conditions for nonconvex optimization problems'', {\it Mathematische Operationsforschung und Statistik. Series Optimization} {\bf 11} (2), 181--219  (1980).
		
\vskip3pt\bibitem{tt9} A.\,K.~Kerimbekov, R.\,J.~Nametkulova, A.\,K.~Kadirimbetova, ''Optimality conditions in the problem of thermal control with integral-differential equation'', {\it Vestnik Irkutsk Gosudarstvennogo Universiteta. Matematika} {\bf 15}, 50--61 (2016).  [in Russian]
		
\vskip3pt\bibitem{tt10} A.\,K.~Kerimbekov, E.\,F.~Abdyldaeva, ''On the solvability of the tracking problem in nonlinear vector optimization of oscillatory processes'', {\it Trudy Instituta Matematiki i Mekhaniki} {\bf 30} (2), 103--115 (2024). [in Russian]
		
\vskip3pt\bibitem{tt11} E.\,N.~Mahmudov, ''Infimal convolution and duality in convex optimal control problems with second order evolution differential inclusions'', {\it Evolution Equations and Control Theory} {\bf 10} (1), 37--59 (2021).
		
\vskip3pt\bibitem{tt12} A.\,T.~Ramazanova, A.\,N.~Abdullozhonova, T.\,K.~Yuldashev, ''Optimal mobile control in inverse problem for Barenblatt--Zheltov--Kochina type fractional order equation'', {\it Lobachevskii Journal of Mathematics} {\bf 46} (1), 488--501 (2025). https://doi.org/10.1134/S1995080224608087
		
\vskip3pt\bibitem{tt13} T.\,K.~Yuldashev, B.\,Y.~Ashirbaev, ''Optimal feedback control problem for a singularly perturbed discrete system'', {\it Lobachevskii Journal of Mathematics} {\bf 44} (2), 661--668 (2023).
		
\vskip3pt\bibitem{tt14} T.\,K.~Yuldashev, A.\,T.~Ramazanova, Zh.\,Zh.~Shermamatov, ''Optimal control problem for a linear pseudoparabolic equation with final condition, degeneration and Gerasimov--Caputo operator'', {\it Azerbaijan Journal of Mathematics} {\bf 15} (1), 257--283 (2025). https://doi.org/10.59849/2218-6816.2025.1.259

\vskip3pt\bibitem{Yuldashev3} T.\,K.~Yuldashev, A.\,K.~Fayziyev, ''Periodic solutions of impulsive system of equations with a nonlinear function under the sign of a second-order differential and maxima'', {\it Lobachevskii Journal of Mathematics} {\bf 46} (2), 658--671 (2025). https://doi.org/10.1134/S1995080225600256}

\end{thebibliography}



\hrule
\vskip3mm
{\small \it Journal of Osh University ''Differential Equations'', 2026, Volume 1, Issue 1, P. 67--77.}
\noindent{\phantom{DOI} }
\vskip3mm
\hrule

\vskip7mm
\noindent{\small \bf UDC 517.977.1}

\vskip3mm
\noindent{\Large \bf Optimal Control Problem for an Impulsive Systems of Functional-Differential Equations with a Nonlinear Function under the Sign of a First-Order Differential}

\vskip5mm

\noindent{\bf Mukhametali K. Mamanov}$^*$
\vskip-1pt
{\small Osh State University,  332, Lenin street, Osh, Kyrgyzstan}
\vskip-1pt
{\small
{\bf E-mail:} mamanovmuhametali@gmail.com, \: https://orcid.org/0009-0006-3204-7585}

\vskip3mm

\noindent{\bf Gofurjon A. Khasanov}
\vskip-1pt
{\small Samarkand State University, University blv, 15, Samarkand, 140104 Ozbekistan}
\vskip-1pt
{\small
{\bf E-mail:} khasanov\_g75@mail.ru, \: https://orcid.org/0009-0007-7357-4557}

\vskip3mm

\noindent{\bf Zukhra A. Madatova}
\vskip-1pt
{\small Tashkent University of Information Technology, Tashkent, Ozbekistan}
\vskip-1pt
{\small
{\bf E-mail:} zuhra\_madatova@mail.ru, \: https://orcid.org/0009-0001-0638-4362}

\vskip5mm

\begin{center}
{\bf Received}: November 03, 2025;  {\bf First on-line published}: January 10, 2026
\end{center}

\vskip5mm

\parbox{146mm}{\noindent\textbf{Abstract.} \small In this paper the optimal control problems for an impulsive systems of functional-differential equations with a nonlinear function under the sign of the first-order differential and with maxima are investigated. The Initial value problem is reduced to a system of nonlinear functional-integral equations in a Banach space $BD\big([0,T],\mathbb{R}^n\big)$.
In the fixed values of control function, by the method of contracting mappings proves the existence and uniqueness of state function for a nonlinear systems of functional-integral equations with maxima. Then using functional of quality, we are built Pontryagin's function and obtain criteria of optimality. Then we are proved existence and uniqueness of control function.}

\vskip3mm

\parbox{146mm}{\noindent\textbf{Key words:} \small \it Optimal control, impulse system with maxima, nonlinear function under the differential sign, rotation of homotopic vector fields, nonzero index of an isolated singular point, existence and uniqueness of a periodic solution.}

\vskip3mm

\parbox{146mm}{\noindent\textbf{MSC 2020:} \small 34B10, 34B15, 34B37, 34H05.}


{\centering
\section*{Introduction. Problem statement}}

The theory of functional differential equations has applications in different sciences (see, \cite{chet}--\cite{ka}).
Functional differential equations containing a nonlinear function under the derivative sign arise when solving problems of nonlinear mechanics and nonlinear optimal control. Of theoretical interest is the study of solutions of functional-differential equations when at given points have a discontinuity of the first kind.

There are a lot of publications of studying on differential equations with impulsive effects, describing many
natural and practical processes, are appeared \cite{t1}--\cite{t7}. The interest on the study of nonlocal
problems for the impulsive differential equations is high by the mathematicians \cite{t8}--\cite{t21}.

Optimal control theory constitutes a strong area within the mathematical sciences, offering a wide range of practical applications. Many real-world problems reduce to finding an optimal control function along with the corresponding state function. A wide range of analytical and numerical methods has been developed to solve such problems effectively across disciplines in science and engineering (see, \cite{tt1}--\cite{tt14}).

In this paper problems of optimal control for an impulsive systems of functional-differential equations with a nonlinear function under the sign of the first-order differential are considered. On the interval $[0,T]$ for $t\ne t_{i} ,\, \, i=1,2,...,p$ an impulsive systems of first-order functional-differential equations are considered
\begin{equation} \label{Grind1}
\frac{d}{dt} \Big[x(t)+H(x(t))\Big]=F\left(t,x(t),\max \big\{x(\tau ):\tau \in [t-h,t]\big\},u(t)\right),
\end{equation}
where $0=t_{0} <t_{1} <...<t_{p} <t_{p+1} =T,$ $\,x\in \bar{X},$ $\,\bar{X}$ is closed set in bounded space $\mathbb{R}^{n} $, $F(t,x,y,u)\in C\left([0,T]\times \mathbb{R}^{n} \times \mathbb{R}^{n}\times\Upsilon,\mathbb{R}^{n} \right),$  $H(x)\in PC\left(\mathbb{R}^{n}, \mathbb{R}^{n} \right),$ $\,0<h={\rm const}.$

The system \eqref{Grind1} is studied under following conditions
\begin{equation} \label{Grind2}
x(0)=x_0\neq 0,\, \, \,\; H(x(0))=H(x_0)=H_0\neq 0,
\end{equation}
\begin{equation} \label{Grind3}
x(\xi )=\varphi (\xi),\, \, \,\; \xi \in \big[-h,0^{-}\big]
\end{equation}
and under conditions with given impulses
\begin{equation} \label{Grind4}
x\left(t_{i}^{+} \right)-x\left(t_{i}^{-} \right)=V_{1,i} \left(x\left(t_{i} \right)\right),\, \, \, i=1,2,...,p,
\end{equation}
\begin{equation} \label{Grind5}
H\left(x\left(t_{i}^{+} \right)\right)-H\left(x\left(t_{i}^{-} \right)\right)=V_{2,i} \left(x\left(t_{i} \right)\right),\, \, \,  i=1,2,...,p,
\end{equation}
where $x(t) \in PC\big([0,T],\mathbb{R}^{n} \big)$ is state function, $u(t) \in C\big([0,T],\mathbb{R}^{n} \big)$ is control function, $V_{\kappa} \in C\big(\mathbb{R}^{n},\mathbb{R}^{n} \big),\, \, \kappa =1,2$, $\,x\left(t_{i}^{+} \right)$, $\,x\left(t_{i}^{-} \right)$ are right and left-hand limits.

We use the Banach space $C\big([0,T],\mathbb{R}^{n} \big)$, which contains a vector function $x(t)$, defined and continuous on the segment $[0,T]$, with the norm
\[\left\| \, x(t)\, \right\| =\sqrt{\sum _{j=1}^{n}{\mathop{\max }\limits_{0\le t\le T}} \left|\, x_{j} (t)\, \right| } .\]
Let us also consider the following Banach space
\[PC\left([0,T],\mathbb{R}^{n} \right)=\Big\{x:[0, T]\to \mathbb{R}^{n} ;\, \, x(t)\in C\left(\left(t_{i} ,t_{i+1} \right],\mathbb{R}^{n} \right),\, \, i=1,...,p\Big\},\]
in which $x\left(t_{i}^{+} \right)$ and $x\left(t_{i}^{-} \right)$ $(i=0,1,...,p)$ exist and are bounded, with norm
\[\left\| \, x(t)\, \right\| _{PC[0,T]} =\max \left\{\left\| \, x(t)\, \right\| _{C[t_{i} ,t_{i+1} ]} ,\, \, \, i=1,2,...,p\right\}.\]

We consider a control function
$$u(t) \in \Upsilon=\big\{ u: |\,u(t)\,| \leq M^{*} , \;t \in [0,T],\; 0<M^{*}={\rm const} \big\}$$
and the following functional of quality
\begin{equation}\label{func}
J[u]=\big[x(T)-\alpha \big]^2+\beta \int \limits_{0}^{T} \varepsilon(t) u^2(t)dt,
\end{equation}
where $\alpha, \beta={\rm const}$, $\varepsilon(t) \in C[0,T]$.

\noindent\textbf{Problem statement.} Find a control function $u^{*}(t)\in \Upsilon$ and corresponding values of state function $x^{*}(t)\in PC\big([0, T],\mathbb{R}^{n} \big)$ providing a minimum of functionality \eqref{func} and for all $t\in [0, T],\, \, t\ne t_{i} ,\, \, \, i=1,2,...,p$ state function satisfies the given system \eqref{Grind1}, the conditions \eqref{Grind2} together with the condition \eqref{Grind3}, for $t=t_{i} ,\, \, i=1,2,...,p\, \, \big(0<t_{1} <t_{2} <...<t_{p} <T\big)$ satisfies the impulse conditions \eqref{Grind4} and \eqref{Grind5}.

{\centering
\section{Reduction of the problem \eqref{Grind1}-\eqref{Grind5} to a system of nonlinear functional integral equations}}

Let the function $x(t)\in PC\big([0,T],\mathbb{R}^{n} \big)$ be a solution of the problem \eqref{Grind1}-\eqref{Grind5} for fixed values of control function. Then, taking into account the equalities $x\big(0^{+}\big)=x(0)$, $\,x\big(t_{p+1}^{-} \big)=x(t)$, integrating the system \eqref{Grind1} once in each of the intervals: $\left(0,t_{1} \right]$, $\,\left(t_{1} ,t_{2} \right]$, $\, .\, .\, .\, $, $\,\left(t_{p} ,t_{p+1} \right]$ and combining these integrals, we obtain:
\[\int \limits_{0}^{t}F\left(s,x(s),\max \big\{x(\tau ):\tau \in \left[s-h,s\right] \big\},u(s)\right)ds \]
\[=\left[x\left(t_{1} \right)-x\left(0^{+} \right)\right]+\left[x\left(t_{2} \right)-x\left(t_{1}^{+} \right)\right]+\,.\,.\,.\,+\left[x(t)-x\left(t_{p}^{+} \right)\right]\]
\[+\left[H(x(t_{1} ))-H(x(0^{+} ))\right]+\left[H(x(t_{2} ))-H(x(t_{1}^{+} ))\right]+\,.\,.\,.\,+\left[H(x(t))-H(x(t_{p}^{+} ))\right]\]
\[=-x(0)-\left[x\left(t_{1}^{+} \right)-x\left(t_{1} \right)\right]-\left[x\left(t_{2}^{+} \right)-x\left(t_{2} \right)\right]-\,.\,.\,.\,-\left[x\left(t_{p}^{+} \right)-x\left(t_{p} \right)\right]+x(t)\]
\[-H(x(0))-\left[H(x(t_{1}^{+} ))-H(x(t_{1} ))\right]-\left[H(x(t_{2}^{+} ))-H(x(t_{2} ))\right]\]
\[-\,.\,.\,.\,-\left[H(x(t_{p}^{+} ))-H(x(t_{p} ))\right]+H(x(t)).\]
Taking into account the impulse conditions \eqref{Grind4} and \eqref{Grind5}, we write the last equalities as follows
\[
x(t)+H(x(t))=x(0)+H(x(0))+\sum _{0<t_{i} <t}\Big[V_{1,i} \left(x\left(t_{i} \right)\right)+V_{2,i} \left(x\left(t_{i} \right)\right)\Big]+
\]
\begin{equation} \label{Grind6}
+\int \limits_{0}^{t}F\left(s,x(s),\max \big\{x(\tau ):\tau \in [s-h,s]\big\},u(s)\right)ds.
\end{equation}
For the function $x(t)\in PC\big([0,T],\mathbb{R}^{n} \big)$ in the representation \eqref{Grind6} we will use the conditions \eqref{Grind2} and obtain a system of nonlinear functional-integral equations (SNFIE):
\[
x(t)=x_{0} -H(x(t))+H_{0}+\int \limits_{0}^{t}F\left(s,x(s),\max \big\{x(\tau):\tau \in \left[s-h,s\right]\big\},u(s)\right)ds+
\]
\begin{equation} \label{Grind10}
+\sum _{0<t_{i} <t}\Big[V_{1,i} \left(x\left(t_{i}\right)\right)+V_{2,i} \left(x\left(t_{i} \right)\right)\Big].
\end{equation}
\begin{lemma} \label{LL1} Let be fulfilled the following conditions:\\
$1).$ $M_{F} =\left\| \, F(t,x,y,u)\, \right\| _{C[0,T]} ,\, \, \, 0<M_{F} ={\rm const}$;\\
$2).$ $M_{V_{\kappa}} ={\mathop{\max }\limits_{1\le i\le p}} \left\| \, V_{\kappa ,i}(x)\, \right\| _{C[0,T]} ,\, \, \, 0<M_{V_{\kappa } } ={\rm const},\, \, \kappa =1,2$;\\
$3).$ $\left|\, H\left(x_{1} \right)-H\left(x_{2} \right)\, \right|\le L_{H} \left|\, x_{1} -x_{2} \, \right|,\, \, \, 0<L_{H} ={\rm const}$.

Then for the SNFIE \eqref{Grind10} is true the following estimate
\begin{equation} \label{Grind11}
\left\| \, x(t)-x_{0} \, \right\|_{PC[0,T]} \le \frac{TM_F+p\big(V_1+V_2\big)}{1+L_{H}}.
\end{equation}
\end{lemma}

\begin{proof}[\bf \indent Proof] We rewrite the SNFIE \eqref{Grind10} as
\[
\|\,x(t)-x_{0}\,\|_{PC[0,T]}+\|\,H(x(t))-H(x_{0})\,\|_{PC[0,T]} \leq
\]
\[\leq \int \limits_{0}^{t}\|\,F\left(s,x(s),\max \big\{x(\tau):\tau \in \left[s-h,s\right] \big\},u(s)\right)\,\|_{C[0,T]}ds +\]
\begin{equation} \label{Grind12}
+\sum _{0<t_{i}<t}\Big[\|\,V_{1,i} \left(x\left(t_{i} \right)\right)\,\|+\|\,V_{2,i} \left(x\left(t_{i} \right)\right)\,\|\Big].
\end{equation}

From \eqref{Grind12} we and obtain that
\[\big(1+L_{H} \big)\left\| \, x(t)-x_{0} \, \right\| _{PC[0,T]}\leq TM_F+p\big(V_1+V_2\big).\]
Hence, we get \eqref{Grind11}. Lemma \ref{LL1} is proved. \end{proof}

\begin{lemma}[\cite{Yuldashev3}] \label{LL2} For the difference of two functions with maxima there holds the following estimate
\[
\left\| \, \max \big\{x(\tau ):\tau \in [t-h,t]\big\}-\max \big\{y(\tau ):\tau \in [t-h,t]\big\}\, \right\| _{C[0,T]}
\]
\[
\le \left\| \, x(t)-y(t)\, \right\| _{C[0,T]} +h\left\| \, x'(t)-y'(t)\, \right\| _{C[0,T]},
\]
where $0<h={\rm const}$.
\end{lemma}

{\centering
\section{Unique solvability of the initial value problem}}

By the $BD\left([0,T],\mathbb{R}^{n} \right)$ we denote the Banach space on the interval $[0,T]$ with the norm
\[\left\| \, x(t)\, \right\| _{BD[0,T]} \le \left\| \, x(t)\, \right\| _{PC[0,T]} +h\, \left\| \, x'(t)\, \right\| _{PC[0,T]},\]
$0<h={\rm const}$.

For fixed values of the control function $u(t)$ we proof the following theorem.
\begin{theorem} \label{TT1} Suppose that the conditions of Lemma \ref{LL1} and the following conditions are satisfied:\\
$1).$ $\varphi (\xi )\in C[-h,0]$;\\
$2).$ $M_{H} =\max \limits_{0\le t\le T} \left|\, H\left(x\right)\, \right|<\infty ,\, \, \, \, M_{H'} =\max \limits_{0\le t\le T} \left|\, \frac{d}{dt}H\left(x(t)\right)\, \right|<\infty $;\\
$3).$ For all $t\in [0,T]$, $x,y \in \mathbb{R}^{n} $ there exist functions $\,0<L_{\kappa ,F} (t)\in C\left([0,T],\mathbb{R}^{n} \right),\, \, \, \, \, \kappa =1,2,$ such that
\[\left|F(t,x_{1} ,y_{1} )-F(t,x_{2} ,y_{2} )\right|\le L_{1,F} (t)\, \left|\, x_{1} -x_{2} \, \right|+L_{2,F} (t)\, \left|\, y_{1} -y_{2} \, \right|;\]
$4).$ For all $x\in \mathbb{R}^{n} ,\, \, i=0,1,...,p$ there exist $0<L_{V_{\kappa ,i}} ={\rm const},\, \, \, \kappa =1,2,$ such that
\[\left|\, V_{\kappa ,i} (x_{1} )-V_{\kappa ,i} (x_{2} )\, \right|\le L_{V_{\kappa ,i}} \left|\, x_{1} -x_{2} \, \right|;\]
$5).$ For all $x\in \mathbb{R}^{n}$, there exists $0<L_{H'} ={\rm const},$ such that
\[\left|\, \frac{d}{dt}\Big[H (x_{1})-H (x_{2})\Big]\, \right|\le L_{H'} \left|\, x_{1} -x_{2} \, \right|;\]
$6).$ The radius of the inscribed ball in $X$ is greater than $ \frac{TM_{F} }{1+L_{H}} +p\frac{M_{V_{1} } +M_{V_{2}}}{1+L_{H}};$\\
$7).$ $\rho =\chi _{11} +h\chi _{21} <1,$ where $\chi _{11} $  and $\chi _{21} $ are determined below from the formulas \eqref{Grind24} and \eqref{Grind27}.

Then the problem \eqref{Grind1}--\eqref{Grind5} for all $t\in [0, T],\, \, t\ne t_{i} , \, \, i=1,2,...,p$ has a unique solution $x(t)\in BD\left([0,T],\mathbb{R}^{n} \right),$ which is found from the system of functional-integral equations \eqref{Grind10}.
\end{theorem}

\begin{proof}[\bf \indent Proof] We will show that the right-hand side of the system of equations \eqref{Grind10} as an operator takes a ball with radius $\frac{TM_{F}}{1+L_{H}} +p\frac{M_{V_{1}} +M_{V_{2}}}{1+L_{H}}$ into itself and is a contraction operator. So, according to the Lemma \ref{LL1}, we have the estimate \eqref{Grind11}:
\[\left\| \, x(t)-x_{0} \, \right\|_{PC[0,T]} \le \frac{TM_{F}}{1+L_{H}} +p\frac{M_{V_{1}} +M_{V_{2}}}{1+L_{H}} .\]
In the future we need the derivative of the system \eqref{Grind10}:
\begin{equation} \label{Grind20}
x'(t)+\frac{d}{dt}H(x(t))=F\big(t,x(t),\max \big\{x(\tau):\tau \in [t-h,t]\big\},u(t)\big).
\end{equation}

Similarly to the Lemma \ref{LL1}, passing to the norm in the equation \eqref{Grind20}, we obtain
\begin{equation} \label{Grind21}
\left\| \, x'(t)\, \right\|_{PC[0,T]} \le \frac{M_{F}}{1+M_{H'}}.
\end{equation}
We consider a difference of two functions $x(t)-y(t),$ where the functions $x(t)$ and $y(t)$ are satisfied the system of equations \eqref{Grind10}. By the conditions of the theorem, from \eqref{Grind10} we have
\[
\left|\, x\left(t\right)-y\left(t\right)\, \right|+ L_{H} \left|\, x\left(t \right)-y\left(t \right)\, \right| \leq
\int \limits_{0}^{t}\Big\{L_{1,F} (s)\left|\, x\left(s \right)-y\left(s \right)\, \right|+
\]
\[ +L_{2,F} (s)\left|\, \max \big\{x(\tau ):\tau \in [s-h,s] \big\}-\max \big\{y(\tau):\tau \in [s-h,s] \big\}\, \right|\Big\}ds+\]
\begin{equation} \label{Grind22}
+\sum _{i=1}^{p}\Big[L_{V_{1,i}} +L_{V_{2,i}} \Big]\big|\, x\left(t_{i} \right)-y\left(t_{i} \right)\, \big|.
\end{equation}
Passing to the norm and applying Lemma \ref{LL2} to the inequality \eqref{Grind22}, we have
\[
\left\| \, x(t)-y(t)\, \right\| _{PC[0,T]} + L_{H} \left\| \, x(t)-y(t)\, \right\| _{PC[0,T]} \leq
\]
\[ \leq T\left[\left(\left\| \, L_{1,F} (t)\, \right\|_{C[0,T]} +\left\| \, L_{2,F} (t)\, \right\|_{C[0,T]} \right)\left\| \, x(t)-y(t,x)\, \right\|_{PC[0,T]}+ \right. \]
\[\left. +h\, \left\| \, L_{2,F} (t)\, \right\|_{C[0,T]} \, \left\| \, x'(t )-y'(t )\, \right\| _{PC} \right]+\]
\begin{equation} \label{Grind23}
+\sum _{i=1}^{p}\left[L_{V_{1,i}} +L_{V_{2,i}} \right]\left\| \, x\left(t \right)-y\left(t \right)\, \right\|_{PC[0,T]} .
\end{equation}
If in \eqref{Grind23} we make designations
\begin{equation} \label{Grind24}
\chi _{11} =\frac{T \left(\left\| \, L_{1,F} (t)\, \right\| _{C[0,T]} +\left\| \, L_{2,F} (t)\, \right\|_{C[0,T]} \right)+\sum _{i=1}^{p}\left[L_{V_{1,i}} +L_{V_{2,i}} \right]}{1+L_{H}},
\end{equation}
\[
\chi _{12} = \frac{T\left\| \, L_{2,F} (t)\, \right\|_{C[0,T]}}{1+L_{H}},
\]
then, by virtue of $\chi _{11}>\chi _{12}$, from \eqref{Grind23} obtain
\begin{equation} \label{Grind25}
\left\| \, x(t)-y(t)\, \right\|_{PC} \leq
\chi _{11} \Big[\left\| \, x(t)-y(t)\, \right\|_{PC} +h \left\| \, x'(t)-y'(t)\, \right\|_{PC} \Big].
\end{equation}

By similar way, according to the assumptions of the theorem, from \eqref{Grind20} we have
\begin{equation} \label{Grind26}
 \left\| \, x'(t)-y'(t)\, \right\| _{PC}
\leq \chi _{21} \left\| \, x(t)-y(t)\, \right\|_{PC} +h\,\chi _{22} \left\| \, x'(t)-y'(t)\, \right\| _{PC},
\end{equation}
where
\begin{equation} \label{Grind27}
\chi _{21} =\frac{\left(\left\| \, L_{1,F} (t)\, \right\|_{C[0,T]} +\left\| \, L_{2,F} (t)\, \right\| _{C[0,T]} \right)
+ \sum _{i=1}^{p}\left[L_{V_{1,i}} +L_{V_{2,i}} \right]}{1+L_{H'}},
\end{equation}
\[\chi _{22} =\frac{\left\| \, L_{2,F} (t)\, \right\| _{C[0,T]}}{1+L_{H'}}.\]
Since $\chi _{21}>\chi _{22}$, multiplying both sides of \eqref{Grind26} to positive constant $h$ and the result adding to \eqref{Grind25}, we obtain
\begin{equation} \label{Grind28}
 \left\| \, x(t)-y(t)\, \right\| _{BD[0,T]} \le \rho \, \left\| \, x(t)-y(t)\, \right\| _{BD[0,T]},
\end{equation}
where $\rho =\chi _{11} +h\,\chi _{21}.$

According to the last condition of the theorem, $\rho <1.$ So, from the estimate \eqref{Grind28} we deduce that the operator on right-hand side of the system \eqref{Grind10} is compressing. From the estimates \eqref{Grind11}, \eqref{Grind21} and \eqref{Grind28} implies that there exists unique fixed point $x(t,x_{0})\in BD\big([0,T], \mathbb{R}^n\big).$ The Theorem \ref{TT1} is proved. \end{proof}

{\centering
\section{Optimal control function}}
	
Let $u^{*} (t)$ be optimal control function:
	\[\Delta \, J \big[u^{*} (t)\big]=J \big[u^{*} (t)+\Delta \, u^{*} (t)\, \big]-J\, \big[u^{*} (t)\big]\ge 0,\]
where  $u^{*} (t)+\Delta \, u^{*} (t)\in C[0,T].$
	
Let us consider equation \eqref{Grind10} and the problem of finding the control function
\[u^{*} (t)\in \Upsilon=\Big\{ u^{*} :\, \left|\, u^{*} (t)\, \right|\le M^{*}_{0},\, \, t\in [0,T] \Big\}\]
and the corresponding state function $x^{*}(t)$ that minimize the functional \eqref{func}.

Let's construct the Pontryagin function:
\[H\left(u (t),x (t)\right)=\psi (t)\Big[ F\left(t,x(t),\max \big\{x(\tau):\tau \in \left[t-h,t\right] \big\},u(t)\right)+\]
\[ +V_{1,i} \left(x\left(t_{i}\right)\right)+V_{2,i} \left(x\left(t_{i} \right)\right)\Big]-\beta \varepsilon (t)\, u^{2} (t),\]
where the function $\psi(t)$ will be determined from the equation:
\begin{equation} \label{Grind29}
\dot{\psi} (t)=-\psi (t)
\end{equation}
with condition
\begin{equation} \label{Grind30}
\psi(T)=-2\big[x(T)-\alpha \big].
\end{equation}
Solving this differential equation (\ref{Grind29}), we obtain
\[\psi (t)=C_{0} \exp \{-t \}.\]
To find $C_{0}$ we use the condition (\ref{Grind30})
\[
\psi (T)=C_{0} \exp \{-T \}=-2\bigg[x_{0} -H(x(T))+H_{0}+\int \limits_{0}^{T}F\left(s,x(s),\max \big\{x(\tau):\tau \in [s-h,s]\big\},u(s)\right)ds+\]
\[+\sum _{0<t_{i} <T}\Big[V_{1,i} \left(x\left(t_{i}\right)\right)+V_{2,i} \left(x\left(t_{i} \right)\right)\Big]-\alpha \bigg].\]
Hence, we get
\[\psi (t)=C_{0} \exp \{-t \}=-2\big(x_0+H_0\big)\exp \{T-t \}-2\exp \{T-t \}\bigg[-H(x(T))+\]
\[+\int \limits_{0}^{T}F\left(s,x(s),\max \big\{x(\tau):\tau \in \left[s-h,s\right] \big\},u(s)\right)ds+\]
\begin{equation} \label{Grind31}
+\sum _{0<t_{i} <T}\Big[V_{1,i} \left(x\left(t_{i}\right)\right)+V_{2,i} \left(x\left(t_{i} \right)\right)\Big]-\alpha \bigg].
\end{equation}

The optimality condition is as follows:
\[H_{u}\left(u (t),x (t)\right)=\]
\begin{equation} \label{Grind32}
=\psi (t) F_{u}\left(t,x(t),\max \big\{x(\tau):\tau \in \left[t-h,t\right] \big\},u(t)\right)-2\beta \varepsilon (t)\, u (t)=0.
\end{equation}

We suppose that $\varepsilon (t) \neq 0$. Substituting expression (\ref{Grind31}) into equation (\ref{Grind32}), we obtain an equation with respect to the control function $u (t)$:
\[u (t)=\bigg\{P(t)+Q(t)\bigg[-H(x(T))+\int \limits_{0}^{T}F\left(s,x(s),\max \big\{x(\tau):\tau \in \left[s-h,s\right] \big\},u(s)\right)ds+\]
\begin{equation} \label{Grind33}
+\sum _{0<t_{i} <T}\Big[V_{1,i} \left(x\left(t_{i}\right)\right)+V_{2,i} \left(x\left(t_{i} \right)\right)\Big] \bigg] \bigg\} F_{u}\left(t,x(t),\max \big\{x(\tau):\tau \in \left[t-h,t\right] \big\},u(t)\right),
\end{equation}
where
\[ P(t)=\frac{-2\big(x_0+H_0\big)\exp \{T-t \}}{2\beta \varepsilon (t)}-\alpha Q(t),\:\:Q(t)=\frac{-2\exp \{T-t \}}{2\beta \varepsilon (t)}.\]

For fixed values of state function $x(t)$ we prove that the following theorem is true.

\begin{theorem}\label{TT2} Let the conditions of the theorem \ref{TT1} are fulfilled. If the following conditions hold\\
$1).$ $M_{F_u} =\max \limits_{0\le t\le T} \left|\, F_u\left(t,x,y,u\right)\, \right|<\infty $;\\
$2).$ For all $t\in [0,T]$, $x,y \in \mathbb{R}^{n} $ there exists function $\,0<L_{F} (t)\in C\left([0,T],\mathbb{R}^{n} \right),$ such that
\[\left|\,F(t,x ,y,u_{1})-F(t,x,y,u_{2})\,\right|\le L_{F} (t)\, \left|\, u_{1} -u_{2} \, \right|;\]
$3).$ For all $t\in [0,T]$, $x,y \in \mathbb{R}^{n} $ there exists function $\,0<L_{F_u} (t)\in C\left([0,T],\mathbb{R}^{n} \right),$ such that
\[\left|\,F_u(t,x ,y,u_{1})-F_u(t,x,y,u_{2})\,\right|\le L_{F_u} (t)\, \left|\, u_{1} -u_{2} \, \right|;\]
$4)$. $\rho_1<1$, where \[\rho_1=M_0L_{F_{u}}+TM_{F}L_{F_{u}}+TM_{F_u}L_{F},\:\;M_0=\|\,P(t)\,\|+\|\,Q(t)\,\|\Big[M_H+p\Big(M_{V_{1}}+M_{V_{2}}\Big) \Big].\]

Then the functional equation (\ref{Grind33}) has a unique solution $u(t) \in C\big([0,T],\mathbb{R}^n\big)$.
\end{theorem}

\begin{proof}[\bf \indent Proof] We use the method of contracting mapping in the space $C\big([0,T],\mathbb{R}^n\big)$.
We derive
\begin{equation} \label{Grind34}
\|\,u (t)\,\|\leq \bigg\{\|\,P(t)\,\|+\|\,Q(t)\,\|\Big[M_H
+TM_F+p\Big(M_{V_{1}}+M_{V_{2}}\Big) \Big] \bigg\} M_{F_{u}}<\infty.
\end{equation}

Further, we use the following scheme:
\[|\,f^kg^k-f^{k-1}g^{k-1}\,|\leq |\,f^kg^k-f^{k-1}g^{k}+f^{k-1}g^{k}-f^{k-1}g^{k-1}\,|\leq \]
\[ \leq |\,g^{k}\,||\,f^k-f^{k-1}\,|+|\,f^{k-1}\,||\,g^k-g^{k-1}\,|.\]
Then we obtain
\[ \|\,u (t)-\vartheta(t)\,\|\leq \bigg\{\|\,P(t)\,\|+\|\,Q(t)\,\|\Big[M_H+p\Big(M_{V_{1}}+M_{V_{2}}\Big) \Big]\bigg\} \times\]
\[\times L_{F_{u}}\|\,u(t)-\vartheta (t)\,\|+TM_{F}L_{F_{u}}\|\,u(t)-\vartheta (t)\,\|+TM_{F_u}L_{F}\|\,u(t)-\vartheta (t)\,\| \leq \]
\begin{equation} \label{Grind35}
\|\,u (t)-\vartheta(t)\,\| \leq \rho_{1} \|\,u (t)-\vartheta(t)\,\|,
\end{equation}
where
\[\rho_1=M_0L_{F_{u}}+TM_{F}L_{F_{u}}+TM_{F_u}L_{F},\:\;M_0=\|\,P(t)\,\|+\|\,Q(t)\,\|\Big[M_H+p\Big(M_{V_{1}}+M_{V_{2}}\Big) \Big].\]

Consequently, from the estimates (\ref{Grind34}) and (\ref{Grind35}) follows that the functional equation (\ref{Grind33}) has a unique solution $u(t) \in C\big([0,T],\mathbb{R}^n\big)$.
\end{proof}

One can solve the equations \eqref{Grind10} and \eqref{Grind33} as a one system and obtain control function $u(t)$ and state function $x(t)$.

{\centering
\section*{Conclusion}}

In the segment  $[0,T]$ we consider a nonlinear functional-differential equation with nonlinear function under the sign of first order differential. The functional-differential equation is considered with initial value conditions. To find control function in nonlinear problem, we use minimization of functional of quality.

For fixed values of control function we solve the nonlinear functional equation with initial conditions. For fixed values of state function we solve the nonlinear functional equation with final-point conditions.

The Pontryagin function is built and the condition of optimality is derived. In poving theorem we use the method of contracting mapping.

Moreover, the results obtained in this work will allow us in the future to investigate direct and inverse optimal control problems for other kind of functional-differential equations of mathematical physics with different quality functionals.


\noindent
\textbf{Data availability} This manuscript has no associated data.

\noindent
\textbf{Ethical Conduct} Not applicable.

\noindent
\textbf{Conflicts of interest}
The authors declare that there is
no conflict of interest.

\renewcommand{\refname}{\begin{center}{\Large\bf References} \end{center}}
\makeatletter
\renewcommand{\@biblabel}[1]{#1.\hfill}
\makeatother
%

\bibliographystyle{amsplain}


\end{document}


